Author's response to reviews

Title: Impact of diabetes on hospital admission and length of stay among a general population aged 45 year or more: a record linkage study.

Authors:

Elizabeth J Comino (E.Comino@unsw.edu.au)
Mark F Harris (M.F.Harris@unsw.edu.au)
Islam Fakhrul (M.Islam@unsw.edu.au)
Duong T Tran (D.Tran@uws.edu.au)
Bin Jalaludin (Bin.Jalaludin@sswahs.nsw.gov.au)
Louisa Jorm (L.Jorm@uws.edu.au)
Jeff Flack (Jeff.Flack@sswahs.nsw.gov.au)
Marion Haas (Marion.Haas@chere.uts.edu.au)

Version: 2
Date: 29 July 2014

Author's response to reviews: see over
21st July 2014

The Editor,
BMC Health Services Research,
BioMed Central

Paper Title: Impact of diabetes on hospital admission and length among a general population aged 45 year or more: a record linkage study.

Investigators: A/Prof Elizabeth Comino, Prof Mark Harris, Mr Md Fakhrul Islam, Dr Duong Thuy Tran, Prof Bin Jalaludin, A/Prof Jeff Flack, Prof Louisa Jorm, Prof Marion Haas

Dear Editor,

Please find attached a revised copy of our paper as requested.

We have made the following changes to the format of the paper:
1. Added continuous line numbering (page numbers were already inserted)
2. Removed the version number from the footnotes
3. Included the tables and caption for the figure into the main word document
4. Removed a number of page breaks

The authors and their contact details have not altered. All have made substantial contribution to this work and have agreed to publication.

Yours Sincerely

A/Prof Elizabeth Comino